Author's response to reviews

Title: Identifying context factors explaining physician's low performance on communication assessment: an explorative study in general practice

Authors:

Geurt Essers (G.Essers@elg.umcn.nl)
Sandra van Dulmen (S.vanDulmen@nivel.nl)
Chris van Weel (C.vanWeel@elg.umcn.nl)
Cees van der Vleuten (C.vanderVleuten@maastrichtuniversity.nl)
Anneke Kramer (A.Kramer@elg.umcn.nl)

Version: 7 Date: 6 December 2011

Author's response to reviews: see over
Dear Editorial production Team,

The following formatting changes have been made:

Title page: "* All authors contributed to the concept and design of the study. GE, SvD and AK carried out the observations, and analyzed and interpreted the data. All authors commented on and approved the final manuscript." has been removed from the title page.

The ethics statement has been replaced to the "Methods" section (p. 6)

References: One redundant reference has been removed and all references are now cited in the right numerical order.

Tables: Within the tables 1 and 2 the shaded text has been removed and the text is highlighted by making it bold.

Appendix: The appendix has been made as an additional file, cited that way (p.6) and added in an "Additional files" section at the end of the manuscript.

Affiliations: Each affiliation is now written in the following format: Department, Institute, (Street), City, (Postal code), Country. (title page)

Tables: The visible vertical lines have been removed from the tables

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Thank you for your comments.

Kindest regards,
Geurt Essers